Dear Prof. Wang,

Thank you very much for your letter and advice. We have revised the manuscript, and would like to re-submit it for your consideration. We have addressed the comments raised by the reviewers, and the amendments are highlighted in yellow in the revised manuscript. Point by point responses to the reviewers’ comments are listed below this letter. In addition, the format of the manuscript has been updated according to the Guidelines and Requirements for Manuscript Revision and the Format for Manuscript Revision.

This manuscript has been edited and proofread by Editage.

We hope that the revised version of the manuscript is now acceptable for publication in your journal.

I look forward to hearing from you soon.

With best wishes,

Yours sincerely,

Benshuo Cai

We would like to express our sincere thanks to the reviewers for the constructive and positive comments.

Replies to Reviewer #1
Specific Comments
The manuscript adds new perspectives in the management of high risk pregnancies. It is clear and good for the readers

Answer: Thank you very much!

Replies to Reviewer #2
Specific Comments
Interesting case report about removal laparoscopic cerclage. Congratulations.

Answer: Thank you very much!

Replies to Science editor
Specific Comments
The manuscript introduced Stitch removal using laparotomy and rival-induced labor in late pregnancy after laparoscopic cervical cerclage. This is an interesting topic, but the introduction part of this paper is not enough, and the conclusion part is missing in the conclusion part of this paper. I think this article has good clinical value.

Answer: Thank you very much! Several sentences have been added in the introduction part, and the conclusion part has been added in the conclusion part.

Replies to Company editor-in-chief:
Specific Comments
I have reviewed the Peer-Review Report, the full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. The title of the manuscript is too long and must be shortened to meet the requirement of the journal (Title: The title should be no more than 18 words). Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. The author(s) must provide the Signed Informed Consent Form(s) or Document(s) of treatment. For example, authors from China should upload the Chinese version of the document, authors from Italy should upload the Italian version of the document, authors from Germany should upload the Deutsch version of the document, and authors from the United States and the United Kingdom should upload the English version of the document, etc.

Answer: Thank you very much! The title has been shortened to meet the requirement of the journal. The original figure document has been provided. Signed Informed Consent Form(s) or Document(s) of treatment has been provided.